Hiroki Yamamoto<sup>1</sup>, Daichi Yanagisawa<sup>2,1,\*</sup>, Claudio Feliciani<sup>2</sup>, and Katsuhiro Nishinari<sup>2,1</sup>

 <sup>1</sup> Department of Aeronautics and Astronautics, School of Engineering, The University of Tokyo, 7-3-1 Hongo, Bunkyo-ku, Tokyo 113-8656, Japan
 <sup>2</sup> Research Center for Advanced Science and Technology, The University of Tokyo, 4-6-1 Komaba, Meguro-ku, Tokyo 153-8904, Japan

E-mail: \*tDaichi@mail.ecc.u-tokyo.ac.jp

February 2019

**Abstract.** This study investigated the body-rotation behavior adopted by pedestrians to avoid others while moving in congested areas. In such scenarios, body orientation often differs from walking direction, e.g., a pedestrian may step sideways.

The deviation between body orientation and walking direction during collision avoidance was studied by quantitatively evaluating the body rotation for counter-flows in narrow corridors. Simple experiments, in which two pedestrians passed each other, were conducted. It was found that pedestrians rotated their bodies when the corridor width was smaller than the sum of the widths of the two pedestrians. This behavior was explained by analyzing the geometry of two ellipses circumscribing each other in a narrow corridor. A preliminary model was developed, and the deviation between the body orientation and walking direction during passing was successfully simulated.

Finally, a cross-flow experiment, which is much more complex and realistic than the passing experiments, was performed; it was confirmed that body rotation behavior is also a critical factor in complex and realistic scenarios.

*Keywords*: pedestrian dynamics; collision avoidance; body rotation; passing; ellipse; cross flow

#### 1. Introduction

Pedestrian dynamics, which is considered as one of the practical topics of active matters [1], has been investigated extensively over the last two decades [2,3] by means of simulations and experiments.

Simple models for bidirectional flow and evacuation were developed in the early stages of research on this topic. The social force [4, 5] and floor field [6] models are representative of force-based models in continuous space and cellular automata models, respectively. They are useful for simulating collective real-world pedestrian phenomena,

such as lane formation in bidirectional flow and the arching phenomenon that occurs at a bottleneck.

Behavioral experiments have also been conducted with real pedestrians to elucidate pedestrian behaviors in crowds and calibrate various model parameters. The results of various circuit experiments have allowed researchers to produce refined fundamental diagrams, showing a relation between density and velocity (flow) [7, 8]. Lane formation was observed in experiments on bidirectional flow [9–13]. A bidirectional-flow experiment with an oblique intersecting angle was also conducted [14] and investigated with macroscopic models [15]. Furthermore, the zipper effect (alternate merging of two lanes into one lane) and the relation between the width of a bottleneck and flow were revealed by bottleneck-flow experiments [16–18].

Recently, more detailed pedestrian characteristics have been considered and experimentally investigated to develop more realistic models. The social force model was extended by introducing a required space for walking [19] and a self-stopping mechanism [20], resulting in the successful reproduction of realistic fundamental diagrams. On the other hand, many elements e.g., collisions between pedestrians [21–23], forces between pedestrians [24,25], advance avoidance by anticipation [26,27], heterogeneity of walking speed [28,29], and aggressiveness in a situation of conflict [30], have been implemented in the floor field model. Moreover, the agent-based approach [31] has allowed researchers to represent the heterogeneous properties of pedestrians, e.g., age [32]. Furthermore, the behaviors of individual pedestrians have been researched in detail to model pedestrians as beyond simple self-driven particles. The effects of stride (step length) and walking tempo have been studied [33,34] and introduced into various models [35]. The decisionmaking process for pedestrians has been investigated through experiments on the inflow process [36,37], which is a counter process of evacuation [38,39].

Although many pedestrian characteristics have been studied and implemented in various models, research on body-rotation behaviors has only been started very recently [40] ‡, and further detailed investigations are required. Here, it should be emphasized that the body rotation, which is the focus of this study, is not the change in the walking direction due to turning but the change in the body orientation along the same walking direction.

The effect of body rotation is especially important in congested situations. Bidirectional-flow simulations often end up with a deadlock, where no pedestrians can move without additional rules. In cellular automata models, two opposing pedestrians stochastically exchange their positions to resolve a deadlock [41–43]. Although no detailed explanations exist on how real pedestrians exchange positions, it can be assumed that position exchange is formed by penetrating into the opposing crowd with body rotation. Even when pedestrian density is not extremely high, pedestrians often employ body rotation to avoid colliding with one another without greatly changing their walking direction. This implies that the degree and frequency of body rotation may be related to

 $\ddagger$  Jin *et al.* observed body rotations in their bidirectional-flow experiment and modeled them with a cellular automaton model.

the pedestrian density. Assuming this to be true, it should be possible to approximate the density with body-rotation data recorded by personal smartphones equipped with gyro sensors [44].

Many models represent pedestrians using circles as they are easy to simulate due to their rotational symmetry. However, this symmetry precludes representing the effect attributed to pedestrian orientation. In other words, circular pedestrians always occupy the same region irrespective of their orientation, which makes it impossible to study the effect of body rotation.

One solution to this problem is "elliptic pedestrians", i.e., pedestrians represented using ellipses. Ellipses allow the difference between the effective shoulder width and effective bust depth during walking to be represented. The use of elliptic pedestrians has been proposed in previous works [45–48]. For example, Chraibi *et al.* [48] utilized elliptic pedestrians in their force-based model and succeeded in reproducing the fundamental diagram of unidirectional flow, which quantitatively agrees well with the experimental data. Herein, the pedestrians were represented with ellipses to allow geometrical differences to be considered; it was confirmed that realistic body rotation was reproduced with elliptic pedestrians.

The focus of this study was originally restricted to a simple case, i.e., collision avoidance between two pedestrians when passing through a narrow corridor. This was considered to be one of the most fundamental situations appropriate for research on body-rotation behavior, since the minimal interaction is collision avoidance between two pedestrians. In-detail investigations of microscopic interactions would contribute to the understanding of macroscopic phenomena. Moussaid *et al.* [49] thoroughly analyzed collision avoidance between two pedestrians when changing walking direction and speed to discuss collective patterns in bidirectional crowd flows.

Although the walking behavior of a single pedestrian has been investigated thoroughly in various studies [50], such studies have not focused on the interactions between pedestrians. There are also many sophisticated works on collision avoidance between two pedestrians; however, the motivations of those works were different from those of ours.

One motivation of these previous works was the investigation of personal space. Gorrini *et al.* [51] performed experiments where two pedestrians approached each other and clarified that the size of the front zone of personal space changes depending on the pedestrians' walking speeds.

The other motivation was studying collision avoidance in a wide area. Olivier *et al.* [52] performed crossing experiments at an angle of 90° and revealed that the latter pedestrian gave way to the former pedestrian by decreasing his/her walking speed. Daamen *et al.* [53] and Huber *et al.* [54] performed crossing experiments at various crossing angles. The former study determined that pedestrians tended to pass each other on the right-hand side when the crossing angle increased. The latter study showed that pedestrians adjusted both the walking direction and speed for acute crossing angles, but only adjusted the walking direction for obtuse crossing angles.

Parisi *et al.* [55] conducted experiments with groups of pedestrians and found abrupt evasive maneuvers to increase with the number of pedestrians. They also revealed that pedestrians preferred to change their walking direction rather than decrease their speed. Furthermore, Zanlungo *et al.* [56] extended the model in [49] by introducing preferred direction in collision avoidance and overtaking, they then compared the results of their simulation with real-world data. Since wide areas were assumed in the above studies, pedestrians were able to avoid conflicts by changing their walking direction and speed without body rotation.

In contrast to these previous studies, this study investigated collision avoidance between two pedestrians in a narrow corridor, where collision avoidance would fail without body rotation. The results of this study may contribute to a better understanding of pedestrian behavior and be a source of inspiration to improve existing models. We would also like to mention that the participants of our experiment included both males and females with various ages, thus, our results are not limited to a specific type of pedestrians.

A cross-flow experiment was conducted to examine the importance of body-rotation behavior in more complex and realistic scenarios. We succeeded to observe body rotations in the experiment and to reveal a condition of body rotation in cross flow.

The remainder of this paper is organized as follows. Section 2 describes the analysis of the geometric condition in which two elliptic pedestrians circumscribe each other while passing through a narrow corridor. Section 3 explains the experimental setup and conditions. Section 4 elaborates upon the analysis of the rotational angles in the experiment from various perspectives and examines the validity of the elliptic representation of pedestrians. Section 5 expounds upon the analysis of the deceleration due to body rotation during passing. Section 6 discusses the developed model and how it simulates passing in a narrow corridor. Sections 7 and 8 describes an experiment and analyses, similar to those described in Secs. 3 and 4, conducted for various types of participants. Section 9 presents the results of the cross-flow experiment. Finally, Sec. 10 provides concluding remarks and discussions.

# 2. Geometric analysis of collision avoidance between elliptic pedestrians with body rotation

# 2.1. Elliptic excluded volume of pedestrians

Elliptic pedestrians were considered, as shown in Fig. 1a, and collision avoidance with body rotation in a narrow corridor was investigated, as shown in Fig. 1b. The semimajor and semi-minor axes of an ellipse are described by parameters a (> 0) and b (> 0), such that 2a and 2b correspond to a pedestrian's effective shoulder width and bust depth, respectively. a > b was assumed, so that the width of a pedestrian decreases with rotation. Note that Wąs *et al.* also modeled the physical excluded volume of a pedestrian using an ellipse [47].





Figure 1: (a) Schematic of an elliptic pedestrian. The constant parameters a and b (a > b) represent the semi-major and semi-minor axes of the ellipse, respectively. (b) Schematic of collision avoidance in passage through a narrow corridor. The top and bottom figures show the scenarios before and during passing, respectively. The parameter W is defined as the corridor width. There are two elliptic pedestrians i and j; their passing rotational angles, i.e., the slope angles of the major axes from the initial directions, are represented as  $\theta_i$  and  $\theta_j \in [0^\circ, 90^\circ]$ , respectively; their rotated widths are described by  $d_i \in [2b_i, 2a_i]$  and  $d_j \in [2b_j, 2a_j]$ , respectively.

It has been reported that the effective excluded volume of a walking pedestrians varies with velocity [46]; the velocity-dependent radii of walking pedestrians in a onedimensional circuit has been considered in [19]. Moreover, Chraibi *et al.* modeled a pedestrian using an ellipse with velocity-dependent axes for a two-dimensional case [48]. Given that the dependency of the axes on velocity is different along the longitudinal (walking direction) and lateral (direction vertical to the walking direction) axes, the semi-major and semi-minor axes were exchanged in their model. In other words, the lateral axis is the semi-major axis when the velocity is small, while the longitudinal axis is the semi-major axis when the velocity is large.

One can model a pedestrian more realistically by introducing velocity-dependent axes; however, this study adopted constant axes for simplicity, since the velocity is low in passage through a narrow corridor, as shown in Fig. 1b; when the velocity is low, the velocity-dependent effect considered in [19, 48] would also diminish. Furthermore, the variation in the lateral axis is mainly attributed to the swaying motion of pedestrians, according to [16, 48]. It can be assumed that pedestrians do not sway in the lateral direction when passing through a narrow corridor, and thus the lateral axis was assumed to be constant in this study.



Figure 2: (a) Width of a rotated ellipse d as a function of  $\theta$ . (b) Derivative of the width of a rotated ellipse d' as a function of  $\theta$ . The parameters in both figures are set as (a, b) = (24.9 cm, 15.5 cm), which were determined by the experimental data in Sec. 4.3.

#### 2.2. Relation between passing rotational angle and corridor width

This subsection explains how the relationship between the rotational angle of a body  $\theta$  and corridor width W was obtained.

First, let's focus on the width d occupied by a single elliptic pedestrian when rotating by  $\theta$ . By analyzing the geometry of an ellipse, the relation between width d and passing rotational angle  $\theta$  was calculated:

$$d(\theta, a, b) = 2\sqrt{a^2 \cos^2 \theta + b^2 \sin^2 \theta}.$$
(1)

The detailed derivation is described in Appendix A. Taking the derivative of d with respect to  $\theta$ :

$$d'(\theta, a, b) = \frac{(b^2 - a^2)\sin 2\theta}{\sqrt{a^2 \cos^2 \theta + b^2 \sin^2 \theta}}.$$
 (2)

The width d and its derivative d' are plotted as functions of  $\theta$  in Fig. 2a and 2b, respectively. The parameters were set to (a, b) = (24.9 cm, 15.5 cm), as determined by the empirical data obtained from the experiment explained in Secs. 3 and 4. It can be seen that the slope of d is always negative; however, the steepness varies with the passing rotational angle  $\theta$  (Fig. 2a); Fig. 2b illustrates this feature more clearly. This result indicates that d is sensitive (insensitive) to the change of  $\theta$  where  $\theta$  is around  $45^{\circ}$  (0° and 90°). Therefore, the width of an elliptic pedestrian does not significantly change with oscillatory rotation during normal walking ( $\theta \approx 0$ ). Note that we use the units cm and deg (°) for lengths and angles throughout this paper.

Next, let's deal with two elliptic pedestrians (i and j) who rotate during passing to avoid each other. The semi-major and semi-minor axes of pedestrian i (j) is  $a_i$   $(a_j)$  and  $b_i$   $(b_j)$ , respectively. It was assumed that these axes rotate by  $\theta_i$  and  $\theta_j$ , respectively, such that the sum of their widths is equal to the corridor width W during passing, as in Fig. 1b. More specifically, it was assumed that the two ellipses circumscribed each other as follows:

$$d_i(\theta_i, a_i, b_i) + d_j(\theta_j, a_j, b_j) = W.$$
(3)

Here, the focus was on the situation in which the inequality  $2(b_i + b_j) \leq W \leq 2(a_i + a_j)$  is satisfied; otherwise, rotation does not work effectively. If  $W < 2(b_i + b_j)$ , it is impossible for two elliptic pedestrians to pass each other even with the maximum rotational angle of 90° rotation. In contrast, if  $W > 2(a_i + a_j)$ , two elliptic pedestrians do not need to rotate to pass each other, i.e.,  $\theta = 0^\circ$ .

In Secs. 4.3 and 8.2, the analytic and experimental results of this study are compared, and the validity of using elliptic pedestrians with (3) for modeling body rotation is examined.

# 3. Setup and conditions of corridor experiment 1

A real experiment was performed to investigate the body rotation and side-stepping behavior of pedestrians from the perspective of collision avoidance in a narrow corridor. The experiment was conducted at the Research Center for Advanced Science and Technology (RCAST), The University of Tokyo, Japan. Two courses were constructed using cardboard boxes, as described in Fig. 3a.



Figure 3: (a) Schematic of experimental setup. Two courses were formed by placing cardboard boxes. Four participants (yellow ellipses) standing at the ends of the courses walked to the other ends upon cue. The passing side was specified to the participants in advance to each trial. In this figure, the starting positions in the left-passing case are depicted. (b) Snapshot of the experiment for W = 60 cm in the left-passing case. It can be observed that the two participants pass each other by rotating their bodies.

Trial No.	Passing side	Pair in Course 1	Pair in Course 2
1, 2	left	(1, 2)	(3, 4)
3, 4	right	(1, 2)	(3, 4)
5, 6	left	(2, 3)	(1, 4)
7, 8	right	(2, 3)	(1, 4)
9, 10	left	(1, 3)	(2, 4)
11, 12	right	(1, 3)	(2, 4)

**Table 1:** Overview of the six types of trial sets for each corridor width W. The numbers in<br/>the parentheses represent the ID numbers of the four participants.

The experiment included four male participants (ID number = 1, 2, 3,and 4), ages ranging from 18 to 25. Initially, the participants were arranged at the end of each course. One trial of the experiment started when the participants were instructed to go. The participants first started to walk, passed each other midway through the course, and exited the course from the other end. Figure 3b shows a snapshot taken during the experiment. The participants were instructed to stand and walk on the left or right side of the course before each trial so that they could pass each other without dithering.

A commercial tablet (Nexus 7 (2013)) equipped with a gyro sensor was strapped between the chest and belly of each participant to record angular velocity. The tablets were strapped on the participants' bodies such that the recorded angular velocities would be identical to the angular velocities of their bodies' rotations. The precision and accuracy of the gyro sensor in the tablets were verified in [57, 58], and both were concluded to be below 1%.

The width W of the two courses were varied as 60, 70, 80, 90, 100, 110, 120, and 140 cm, and the angular velocity and travel time of each participant were observed. By changing the passing side and participant pairing, six types of trial sets were executed twice for each W, as summarized in Tab. 1. Two pairs of participants (four participants) were included in each trial. Thus, 2 (pairs)  $\times$  2 (trials)  $\times$  6 (trial sets) = 24 datasets were obtained for each W and 24  $\times$  8 (widths) = 192 datasets in total.

#### 4. Analysis of passing rotational angles in corridor experiment 1

#### 4.1. Time series of angular velocities and rotational angles

In the experiment, the time series of angular velocities  $\omega_i(t)$  (i = 1, 2, 3, and 4 (ID number)) were obtained at approximately 50 Hz from the tablets. Figure 4 shows  $\omega_1(t)$ ,  $\omega_2(t)$ ,  $\omega_3(t)$ , and  $\omega_4(t)$  in one trial with W = 60 cm and left passing. The positive and negative angular velocities correspond to the counter-clockwise and clockwise rotations, respectively. In this example, participants 1 and 2 (3 and 4) were paired together. It can be seen that the magnitudes of the angular velocities started to increase at around t = 2.0 s; this indicates the beginning of rotation.



Figure 4: Time series of angular velocities  $\omega_i(t)$  [deg/s] of the four participants obtained from the tablets in one trial for W = 60 cm, Trial No. = 1 (left passing).  $\omega_i(t) > 0$  ( $\omega_i(t) < 0$ ) corresponds to the counter-clockwise (clockwise) rotation of the participants.

In the left-passing case, the participants faced outward with respect to the course with positive (counter-clockwise) rotation and faced inward with respect to the course (the other participant) with negative (clockwise) rotation. Thus, participants 1, 2, and 3 faced inward with respect to the course, while only participant 4 faced outward with respect to the course, as shown from the data corresponding to around t = 2.5 s in Fig. 4. Note that the direction in which the participants faced during passing were maintained throughout the experiment. Furthermore, no significant difference was found between the left-passing and right-passing cases. Thus, we do not distinguish between the two cases in the following.

By numerically integrating the datasets of angular velocities  $\omega_i(t)$ , the time series of the rotational angle of each participant  $\varphi_i(t) \in [-180^\circ, 180^\circ]$  was obtained as follows:

$$\varphi_i(t) = \int_0^t \omega_i(t) dt \approx \sum_n \omega_{i,n} \delta t_{i,n}, \tag{4}$$

where  $\omega_{i,n}$  and  $\delta t_{i,n}$  are participant *i*'s *n*-th angular velocity and time interval, respectively.

The time interval  $\delta t_{i,n}$  should be small enough, i.e., the measurement frequency of the sensors should be large enough, to obtain accurate rotational angles. It was confirmed that 50 Hz ( $\delta t_{i,n} = 0.02$  s) is large enough; however, due to a bug in the tablets' operating system, the measurement frequency sometimes declined ( $\delta t_{i,n}$  increased). For such time series, it was impossible to compute the accurate rotational angles; therefore, the time series of the pair that included  $\delta t_{i,n} \ge 0.1$  s were discarded. Eventually, two datasets were removed from W = 60, 90, 140 cm, and one dataset was removed from W = 80, 100 cm.

The  $\varphi_i(t)$  of one pair in one trial was plotted for each W as shown in Fig. 5. It can be seen that the participants clearly rotated their bodies when passing for W = 60, 70,

80, and 90 cm, whereas they did not for W = 100, 110, 120, and 140 cm. Actually, a few data points indicated clear rotation for W = 100 cm. Thus, in the following, the focus is placed on W = 60, 70, 80, 90, and 100 cm, where body rotation was observed during passing. Note that deceleration also occurred from 100 cm. See Sec. 5 for details.

# 4.2. Extraction of passing rotational angles

This subsection describes the derivation of the experimental passing rotational angles  $\theta_i \in [0, 90^\circ]$  from the time series of the rotational angles  $\varphi_i$ . To determine the passing rotational angle without ambiguity, it was assumed that the maximum absolute rotational angle  $|\varphi|_{\max,i}$  was the passing rotational angle  $\theta_i$ , i.e.,  $\theta_i = |\varphi|_{\max,i}$ . Note that all passing rotational angles larger than 90° were replaced with 90°.

However, the method has its limitations. First, the oscillation of the rotational angle in normal walking is occasionally regarded as the passing rotational angle. Second, in a few datasets, there were great discrepancies between the times at which each participant achieves the maximum rotational angle, making it inappropriate to regard them as sets of passing rotational angles. Therefore, two thresholds were introduced, one for the passing rotational angle  $\theta_i$  and another for the time gap between the points at which the two participants achieved the maximum rotational angles, which are denoted as  $\Delta t_{|\varphi|_{\text{max}}}$ . Any dataset that did not satisfy the conditions of the two thresholds were converted appropriately, as described in Appendix B, before being used for analysis.

Figure 6a shows the averages and sample standard deviations of  $\Delta t_{|\varphi|_{\text{max}}}$  as functions of W before conversion. From this figure, it can be seen that the values are small for W = 60, 70, and 80 cm, increasing slightly for W = 90 cm, and increasing greatly for W = 100 cm. Thus, a few conversions are necessary for W = 90 cm and many conversions are necessary for W = 100 cm.

Actually, a large  $\Delta t_{|\varphi|_{\text{max}}}$  was obtained when no clear maximum rotational angle was observed, as shown in Figs. 5e-5h. Therefore,  $\Delta t_{|\varphi|_{\text{max}}}$  can be used for judging not only whether the assumption  $\theta_i = |\varphi|_{\text{max},i}$  is valid but also whether collision avoidance occurs with body rotation. Judgment based on the values of the time gap is much easier than that based on the plots of rotational angle evolution, as shown in Fig. 5. Thus, the time gap  $\Delta t_{|\varphi|_{\text{max}}}$  was considered to be a simple and useful indicator for detecting collision avoidance with body rotation.

Figure 6b shows the pairs of passing rotational angles  $(\theta_i, \theta_j)$   $(i \neq j)$  in the experimental data after conversion. It can be seen that the passing rotational angles increased with decreasing corridor width. This result is intuitively agreeable. Moreover, the passing rotational angles were not always equal within a given pair, i.e.,  $\theta_i \neq \theta_j$  in general. Therefore, the rotation of a pair, as opposed to that of an individual, was considered.



Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 11

Figure 5: Time series of the rotational angle  $\varphi_i(t)$  [deg] of one pair in one trial. (a) W = 60 cm, left-passing. (b) W = 70 cm, right-passing. (c) W = 80 cm, left-passing. (d) W = 90 cm, right-passing. (e) W = 100 cm, left-passing. (f) W = 110 cm, left-passing. (g) W = 120 cm, left-passing. (h) W = 140 cm, left-passing. Note that t = 0 in the figures does not always correspond to the beginnings of the trials because 4.5 s data was extracted around passing for visibility.



Figure 6: (a) Average time gap  $\Delta t_{|\varphi|_{\max}}$  between the achievement of the maximum rotational angles  $|\varphi|_{\max,i}$  and  $|\varphi|_{\max,j}$   $(i, j = 1, 2, 3, 4, i \neq j)$  before conversion with sample standard deviation. (b) Pairs of passing rotational angles  $(\theta_i, \theta_j)$  obtained from the experimental data after conversion for various values of W. The black crosses (+), red crosses  $(\times)$ , green triangles  $(\Delta)$ , blue circles  $(\circ)$ , and gray squares  $(\Box)$ represent the experimental data for W = 60, 70, 80, 90, and 100 cm, respectively. Note that the experimental data were plotted such that  $\theta_i \geq \theta_j$  is satisfied.

#### 4.3. Validation of elliptic pedestrians

To validate elliptic pedestrians and assumption (3) during passing, the parameters of elliptic pedestrians were determined using the experimental data.

As discussed in Sec. 4.2 and shown in Fig. 6b, the passing rotational angles of the two participants were generally different. Therefore, it was necessary to investigate the relationship between  $(\theta_i, \theta_j)$  and W. Hence, the relationship between  $d_i + d_j$  (the sum of the widths of the two ellipses) and W was studied; this is much more appropriate in terms of comparison, because  $d_i + d_j$  is one-dimensional, and its physical dimension is equal to that of W.

Since the participants were all male adolescents with similar builds, the semimajor and semi-minor axes of the elliptic pedestrians were assumed to be identical, i.e.,  $a_i = a$  and  $b_i = b$  for i = 1, 2, 3, and 4. Then, (1) and (3) were used along with the experimental passing rotational angles and the semi-major and semi-minor axes of the elliptic pedestrians (a and b) were determined to minimize the following function (the least squares method):

$$f(a,b) = \sum_{\text{all data}} \left[ W - d_i(\theta_i, a, b) - d_j(\theta_j, a, b) \right]^2.$$
(5)

As a result, the parameters (a, b) = (24.9 cm, 15.5 cm) were obtained. Then, the mean and sample standard deviation of  $d_i + d_j$  for each W were computed by substituting



Figure 7: Sum of widths of the two ellipses  $d_i + d_j$  as a function of corridor width W. The green diagonal line represents the theoretical assumption (3), i.e.,  $d_i + d_j = W$ . The red markers are the average values calculated using the least squares method by exploiting the experimental data and (1). The error bars show the sample standard deviations.

the calibrated semi-major and semi-minor axes (a, b) and the experimental passing rotational angles  $(\theta_i, \theta_j)$  into (1). Figure 7 shows the mean and sample standard deviation of  $d_i + d_j$  with the red markers and error bars, respectively.

In Sec. 2, it was assumed that pedestrians passing each other satisfied (3), the green diagonal line in Fig. 7. Thus, the correspondence between the red markers and green diagonal line indicates the validity of this assumption. It can be observed that the markers agree favorably with the diagonal line, namely, the error bars intersect with the green diagonal line except when W = 60 cm.

The discrepancy at W = 60 cm is considered to occur as follows. The walls of the experimental courses were constructed out of cardboard boxes, whose heights (62 cm) were much shorter than those of the participants (Fig. 3a), so that the participants' upper bodies were allowed to pass over the corridor. Therefore, it was considered that the participants pushed their shoulders out of the corridor and passed each other when the corridor width was narrow (W = 60 cm) and collision avoidance was not easy; this decreased the passing rotational angles and increased  $d_i + d_j$ , as can be seen at W = 60 cm in the plot of Fig. 7.

The validity of the parameters (a, b) = (24.9 cm, 15.5 cm) determined from the experimental data was also examined. Since (a, b) are the semi-major and semi-minor axes of the ellipse, respectively, the effective shoulder width and bust depth of a model pedestrian correspond to 2a = 49.8 cm and 2b = 31.0 cm, respectively. In the passing scenario of the experiment, the shoulders of the participants did not come in contact

with one other. This is ascribed to the fact that pedestrians tend to maintain some distance from other pedestrians to avoid physical contact. Therefore, it can be assumed that the physical size of the participants is smaller than the effective size obtained from the experimental data. It has been reported that the average shoulder width and bust depth of Japanese men of age 20-24 years (without clothes) are 44.9 cm and 20.1 cm, respectively [59]. Considering clothing thickness and the fact the participants did not come into contact with each other, the statistical value of shoulder width can be considered to be appropriately smaller than the calibrated value. However, the statistical value of the bust depth was approximately 10 cm smaller than the calibrated value. This is likely because pedestrians are more hesitant about chest-to-chest contact than shoulder-to-shoulder contact. Hence, the calibrated size of the elliptic pedestrians was concluded to be appropriate representations of the effective size of real pedestrians.

# 5. Analysis of deceleration during passing in corridor experiment 1

This section elaborates upon the deceleration effect during passing. Figure 8 shows the average time for traveling 2 m ( $T_{\text{travel}}$ ) as a function of W in course 1 of corridor experiment 1. It can be seen that the qualitative characteristic of the plot changed greatly at W = 100 cm, similar to the case of body rotation. Specifically, the travel time was small and constant for W = 100-140 cm, for which few clear body rotations were observed. By contrast, the travel time increased as W decreased from 100 to 60 cm, for which the participants clearly rotated their body to pass each other. This increase in the travel time is due to deceleration during passing with body rotation.

The travel time was considered to be represented by the sum of the travel time without deceleration  $T_0$  and the increment due to deceleration during passing  $\Delta T$ , as follows:

$$T_{\text{travel}} = T_0 + \Delta T(W), \tag{6}$$

where the explicit formulation of  $\Delta T(W)$  was assumed as follows:

$$\Delta T(W) = \begin{cases} c_1 (W_{\rm cr} - W)^{c_2} & (W \le W_{\rm cr}), \\ 0 & (W \ge W_{\rm cr}). \end{cases}$$
(7)

 $T_0 = 1.29$  s was calculated from the average travel time for W = 100-140 cm, and  $W_{\rm cr}$  was set to  $W_{\rm cr} = 100$  cm. Then, the parameters  $c_1$  and  $c_2$  were determined with the least squares method by using the average travel time for W = 60-100 cm. As a result,  $c_1 = 1.94 \times 10^{-4}$  s/cm<sup> $c_2$ </sup> and  $c_2 = 2.21$  were obtained. The green curve in Fig. 8 is depicted with (6) and (7) and agrees well with the experimental data.

 $\Delta T(W)$  represents the increment of travel time during one pass with body rotation. Thus, the influence of multiple (N times) passes can be estimated as  $\Delta T(W) \cdot N$ , although more data are required to improve the quality of the result for real-world applications.



Figure 8: Average travel time for 2 m ( $T_{\text{travel}}$ ) as a function of corridor width (W) in course 1 of corridor experiment 1. The red markers and error bars represent the averages and sample standard deviations of the experimental results, respectively. The green curve is depicted with (6) and (7), whose parameters were calibrated with the experimental data.

# 6. Simulation model and comparison between experimental and simulation results

#### 6.1. Simulation model for pedestrians with body rotation in a narrow corridor

As described in Sec. 1, pedestrians were represented by circles in many conventional models. There are some models that represent pedestrians with ellipses and other morerealistic shapes; however, the body orientation and walking direction of a pedestrian are considered to be the same in these models. Therefore, these previous models cannot reproduce collision avoidance by body rotation. Thus, a new model was developed to simulate pedestrians' collision-avoidance behavior by body rotation.

Figure 9 shows a schematic of the model, the setup of which is equivalent to that of corridor experiment 1 (explained in Sec. 3). The length and the width of the corridor are L and W, respectively. There is a pedestrian R (L) at the left (right) end of the corridor who moves to the right (left) end in the simulation.

Pedestrians are represented with the same ellipses whose semi-major and semiminor axes are a and b, respectively, and their desired velocity is  $v_i^{\text{des}}$  ( $i \in \{\text{R}, L\}$ ). The position of the center and the rotational angle of pedestrians are described by  $(x_i, y_i)$ and  $\varphi_i$ , respectively.

When the distance between the two pedestrians is larger than the threshold distance, i.e.,  $x_{\rm L} - x_{\rm R} > d_{\rm th}$ , the two pedestrians do not interact with each other and keep walking with their desired velocities  $v_i^{\rm des}$ . When the two pedestrians approach each

Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 16



Figure 9: Schematic of the simulation model. Pedestrians R and L try to decrease the overlap length l by evading and rotating according to (9) and (10), respectively.

other, i.e.,  $x_{\rm L} - x_{\rm R} \leq d_{\rm th}$ , they start to interact.

The pedestrians were considered to evade in the perpendicular direction (y) to their moving direction (x) and rotate their bodies to decrease the overlap length l and avoid each other, as shown in Fig. 9. Thus, pedestrians gradually control their positions  $y_i$ and rotational angles  $\varphi_i$  proportional to the overlap length l, which is calculated by exploiting (1). Furthermore, pedestrians' walking speeds become smaller during sidestepping. Therefore, the equations of motion of pedestrians before passing are described as follows:

$$\frac{dx_i}{dt} = v_i^{\text{des}} \cos(\varphi_i),\tag{8}$$

$$\frac{dy_i}{dt} = k_y^{\rm A} l \cdot \operatorname{sign}(y_i - y_j), \tag{9}$$

$$\frac{d\varphi_i}{dt} = k_{\varphi}^{\rm A}l,\tag{10}$$

where  $k_y^A$  and  $k_{\varphi}^A$  are sensitivity parameters for the overlap length, sign(z) gives the sign of the argument z, and  $y_j$  ( $j \in \{R, L\}$ ) is the position of the opposing pedestrian.

After passing, i.e.,  $x_{\rm L} - x_{\rm R} < -2b$ , pedestrians try to restore their attitudes and move as fast as possible. It was assumed that the restoring behavior was proportional to the evading distance and rotational angle, so that the equations of motion of pedestrians after passing can be described as follows:

$$\frac{dx_i}{dt} = v_i^{\text{des}} \cos(\varphi_i),\tag{11}$$

$$\frac{dy_i}{dt} = -k_y^{\mathrm{R}}(y_i - y_i^0), \qquad (12)$$

$$\frac{d\varphi_i}{dt} = -k_{\varphi}^{\mathrm{R}}(\varphi_i - \varphi_i^0), \qquad (13)$$

where  $k_y^{\text{R}}$  and  $k_{\varphi}^{\text{R}}$  are sensitivity parameters for the deviation from the initial position  $(y_i^0)$  and initial rotational angle  $(\varphi_i^0)$ , respectively.

The origin of the coordinate system was set to the center of the corridor and the x and y axes and positive direction of  $\varphi$  were defined as in Fig. 9. At the beginning of the simulation, pedestrians stood to touch the wall, so that the initial positions and rotational angles of the pedestrians were  $(x_{\rm R}^0, y_{\rm R}^0, \varphi_{\rm R}^0) = (-L/2, W/2 - a, 0)$  and  $(x_{\rm L}^0, y_{\rm L}^0, \varphi_{\rm L}^0) = (L/2, -W/2 + a, 0)$ , respectively. Since the pedestrians R and L moves to the right and left, respectively,  $v_{\rm R}^{\rm des} > 0$  and  $v_{\rm L}^{\rm des} < 0$ .

# 6.2. Comparison of passing rotational angle and travel time between experiment and simulation

The semi-major and semi-minor axes of the pedestrians were set as (a, b) = (24.9 cm, 15.5 cm), as obtained from the analysis shown in Sec. 4. Then, the desired velocity and the threshold distance were determined as  $v_i^{\text{des}} = 155 \text{ cm/s}$ ,  $d_{\text{th}} = 150 \text{ cm}$  from the travel times in the case W = 100, 110, 120, and 140 cm and video data, respectively. To calibrate the parameters of evasion and rotation  $(k_y^A, k_\varphi^A, k_y^R, \text{ and } k_\varphi^R)$ , it was necessary to consider three requirements. First, the passing rotational angles of the experiment discussed in Sec. 4 and those of the simulation should coincide with each other. Second, the travel times of the experiment discussed in Sec. 5 and those of the simulation should also coincide with each other. Finally, the overlap length during passing should be minimized. These three requirements are inconsistent. For example, if the error in travel time is decreased, the overlap length during passing may increase. In this paper, we first tried to minimize the error of the passing rotational angle and travel time between the experiment and simulation, and then confirmed that the overlap length during passing was small enough.

An error function related to the passing rotational angle and travel time was considered:

$$\sum_{W \in W_{\text{exp}}} \left[ (\theta_{\text{sim}}'(W) - \theta_{\text{exp}}'(W))^2 + (T_{\text{travel,sim}}'(W) - T_{\text{travel,exp}}'(W))^2 \right], \quad (14)$$

where  $W_{\text{exp}} = \{60, 70, 80, 90, 100, 110, 120, 140\}$  cm and  $X'_{\text{Y}}$  ( $X = \theta, T_{\text{travel}}, \text{Y} = \exp$ , sim) is the scaled passing rotational angle and scaled travel time, respectively, of the experiment and simulation.  $\theta_{\text{exp}}$  is the average of the plots in Fig. 6b and  $T_{\text{travel,exp}}$  is the data plotted in Fig. 8. Scaling was performed by the following formulation:

$$X'_{\rm Y} = \frac{X_{\rm Y} - X_{\rm exp,min}}{X_{\rm exp,max} - X_{\rm exp,min}}.$$
(15)

Then, the parameter set that minimizes the error function from  $k_y^{\text{A}}$ ,  $k_{\varphi}^{\text{A}}$ ,  $k_y^{\text{R}}$ ,  $k_{\varphi}^{\text{R}}$ , k

Figure 10a shows the passing rotational angle  $\theta$  in the experiment and simulation as functions of the corridor width W. The data plotted in Fig. 6(b) were averaged and used for the experimental results of W = 60, 70, 80, 90, and 100 cm. Similar data

Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 18



Figure 10: (a) Passing rotational angles  $\theta$  (b) travel time for 2 m ( $T_{\text{travel}}$ ) in the experiment and the simulation as functions of the corridor width W. The red markers and error bars represent the averages and sample standard deviations of the experimental results, respectively. The blue circles represent the simulation results. The parameters in the simulation are  $(a,b) = (24.9 \text{ cm}, 15.5 \text{ cm}), (v_{\text{R}}^{\text{des}}, v_{\text{L}}^{\text{des}}) = (155 \text{ cm/s}, -155 \text{ cm/s}), \text{ and } (k_y^{\text{A}}, k_{\varphi}^{\text{A}}, k_y^{\text{R}}, k_{\varphi}^{\text{R}}) = (9.0 \text{ cm/(cm \cdot s)}, 6.0 \text{ deg/(deg \cdot s)}, 5.0 \text{ cm/(cm \cdot s)}, 7.0 \text{ deg/(deg \cdot s)})$ . Equations (8)-(13) were numerically calculated by the Euler method with  $\Delta t = 0.01 \text{ s in the simulation}$ .

were also used for W = 110, 120, and 140 cm. The maximum  $\varphi$  were plotted for the simulation result.

It can be observed that the simulation successfully reproduced the experimental result; however, the error is relatively large at W = 100 cm. This is due to the oscillation during walking in the experiment. The pedestrians in the simulation do not oscillate during walking, so that  $\varphi = 0$  throughout the simulation if  $W \ge 2 \times 2a$  (=99.6) cm.

Note that the minimum corridor width in the simulation is not W = 60 cm but W = 62 cm, since the sum of the minimum widths of two pedestrians is  $2b \times 2 = 62.0$  cm in the simulation. As discussed in Sec. 4.3, the pedestrians could push their shoulders out of the corridor due to the short wall in the experiment. However, such behavior is not considered in the simulation, and thus it is very difficult for the pedestrians to pass each other in the corridor of W = 60 cm in the simulation.

Figure 10b shows the travel times  $T_{\text{travel}}$  in the experiment (same as in Fig. 8) and simulation as functions of the corridor width W. It can be seen that the two results are in agreement.

The above discussion confirms that the proposed model can simulate pedestrians passing in a narrow corridor in terms of passing rotational angle, travel time, and overlap



Figure 11: Speed-headway relation in unidirectional flow. The magenta circles and green plot represent the result of the experiments in [60–62] and the calibrated desired-speed function  $|v^{\text{des}}| = s(h)$  (16), respectively. The parameters in s(h) are set as  $s^{\text{max}} = 139 \text{ cm/s}, h_0 = 49 \text{ cm}, \text{ and } h_1 = 146 \text{ cm}.$ 

length during passing.

#### 6.3. Bidirectional-flow simulation and fundamental diagram

In this subsection, we conducted bidirectional-flow simulations with more than two pedestrians in a narrow corridor and depicted fundamental diagrams to validate the proposed model for crowd simulation. To achieve this, the speed-headway relation should be modeled in addition to our body-rotation model. We assumed that the desired-speed  $|v^{\text{des}}|$  § is given by the following piecewise-linear function:

$$|v^{\text{des}}| = s(h) = \begin{cases} 0 & (0 \le h \le h_0), \\ s^{\max} \frac{h - h_0}{h_1 - h_0} & (h_0 \le h \le h_1), \\ s^{\max} & (h_1 \le h), \end{cases}$$
(16)

where the argument h is the headway distance. The maximum speed  $s^{\text{max}}$  [cm/s],  $h_1$  [cm], and  $h_2$  [cm] are the parameters of the desired-speed function. We exploited the experimental data of unidirectional flow in [60–62] and determined these three parameters through the least squares method. The results were  $s_{\text{max}} = 139 \text{ cm/s}$ ,  $h_0 = 49 \text{ cm}$ , and  $h_1 = 146 \text{ cm}$ . Figure 11 shows the plots of the experimental data and the

<sup>§</sup> The word "desired-speed" is usually used to describe the maximum speed without any obstruction; however, in this context, we used it as the speed determined by the headway distance without body rotation to maintain the consistency of the meaning of the word "desired" from the previous sections.

Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 20



Figure 12: Fundamental diagrams. (a) Speed-density relation. (b) Flow-density relation. The magenta circles, green dashed curve, orange triangles, and blue solid curve represent the result of the unidirectional-flow experiment, unidirectional-flow simulation, bidirectional-flow experiment, and bidirectional-flow simulation, respectively. The corridor length L = 1000 cm, and the corridor width W = 50 and 80 cm in the unidirectional and bidirectional-flow simulations, respectively. The maximum speed  $s^{\max} = 139$  cm/s. Other parameters are the same as in Figs. 10 and 11. Each simulation curve agrees well with the corresponding experimental plot.

desired-speed function (16), which were confirmed to agree well with each other  $(R^2 = 0.73)$ .

By using the calibrated desired-speed function (16), we performed the unidirectional-flow simulations in the periodic corridor (circuit), whose length and width were L = 1000 cm and W = 50 cm, respectively. We controlled the number of pedestrians N from 1 to 17 by 1 and positioned them at equal intervals in the corridor at the beginning of the simulations. Overtaking was not considered, therefore, evading or body-rotational behaviors were not observed. Thus, the desired-speed function (16) dominated the system. As we can see from Fig. 12, the experimental and simulation results agree well each other in both the speed-density (Fig. 12a) and flow-density (Fig. 12b) relations as well as the speed-headway relation (Fig. 11).

Then, we conducted bidirectional-flow simulations. The width of the corridor was set as W = 80 cm, and the number of total pedestrians N was controlled from 2 to 24 by 2. The number of the right- and left-going pedestrians were same (N/2). Both types of pedestrians were positioned at equal intervals in the corridor at the beginning of the simulations. The sum of the shoulder widths of the two pedestrians  $2 \times 2a = 99.6$ cm was larger than W = 80 cm, therefore, the pedestrians needed to evade others and rotate to pass each other. The result of the bidirectional-flow simulations are presented with those of experiments in [60, 61, 63] in Fig. 12. We can see that the simulation results are in agreement with the experimental results.

By comparing the experimental results of the unidirectional and bidirectional flows, we can observe that unidirectional flow achieve larger values of speed and flow when the density is smaller than the critical density  $\approx 2.3 \text{ m}^{-2}$  than bidirectional flow. When the density is greater than the critical density, the values of bidirectional flow become larger than those of unidirectional flow. These characteristic phenomena were successfully reproduced in our simulation results.

In the low-density situation, pedestrians can move freely in unidirectional flow; however, they have to interact with opponent pedestrians to avoid them in bidirectional flow. Thus, speed and flow are greater in unidirectional flow than in bidirectional flow.

In the high-density situation, interactions with other pedestrians are unavoidable in both flows. Overtaking is difficult in unidirectional flow because pedestrians cannot see behind themselves. It is difficult for them to give way to fast followers. Thus, the fundamental diagrams are mainly dominated by the simple speed-headway relation. In contrast, pedestrians can see opponent pedestrians, give way, and pass by each other through evasion and rotation in bidirectional flow. Due to these avoidance behaviors, speed and flow of bidirectional flow remain larger than those of unidirectional flow in the high-density situation.

Our simulation model succeeded reproducing this phenomenon by introducing avoiding behaviors, i.e., evasion to the perpendicular direction and body-rotation. Further, we would like to mention that the introduction of evasion alone was insufficient for bidirectional-flow simulations in a narrow corridor. Since the sum of the shoulder widths of the two pedestrians  $2 \times 2a = 99.6$  cm was larger than the corridor width W = 80 cm, body rotations were necessary to avoid deadlocks.

# 6.4. Computational time

When simulation models are extended to improve their accuracy, more computational time is often required. Therefore, weight between modeling accuracy and computational time is important to enable the practical use of simulation models. In this subsection, we investigated the computational time of bidirectional-flow simulations in a narrow corridor.

We compared four cases. The first case is the same as the proposed model introduced in the former subsections, which includes both evasion in the perpendicular direction and body rotation (Evasion and rotation). In the second case, only evading behavior was considered, i.e., the body-rotation effect was removed (Only evasion). In the third case, only body-rotation behavior was considered, i.e., the evading effect was removed (Only rotation). In the fourth case, both evasion and body-rotation effects were removed, i.e., pedestrians did not avoid each other (No avoidance).

In Evasion and rotation and Only rotation cases, pedestrians have to be represented



Figure 13: Computational times of the simulations of bidirectional flow as functions of the total number of pedestrians N. The red dashed-dotted, green solid, magenta dashed, and blue dotted curves represent the mean of Evasion and Rotation, Only evasion, Only, rotation, and No avoidance cases, respectively. The filled area of each plot represents the sample standard deviation. 100 simulations were conducted for each N. The parameters are the same as those in Fig. 12. The desktop computer with Windows 8.1 Enterprise, Intel(R) Core(TM) i7-4770 CPU @ 3.40GHz, and RAM 24.0 GB was used for the simulations.

by ellipses. On the contrary, in Only evasion and No avoidance cases, there are no effects attributed to elliptic pedestrians. In the first three cases (except No avoidance case), there are interactions between pedestrians, i.e., pedestrians detect other pedestrians (detecting method in our simulation works).

Note that we conducted the simulations in this subsection only for evaluating the computational time. Thus, the latter three cases (except Evasion and rotation) failed to complete accurate simulation, i.e., pedestrians passed through each other in spite of the large overlap lengths.

Figure 13 shows the computational times of the four cases. We see that the computational times increase as the number of pedestrians N increases. This type of power-law like increase of computational time is common for force-based pedestrian models. As we can imagine, the computational time of Evasion and rotation case is the largest, while that of No avoidance case is the smallest. The computational times of Only evasion and Only rotation cases are similar and closer to that of Evasion and rotation case than No avoidance case. This result indicates that introduction of body-rotation behavior to the simulation models does increase the computational time; however, the increment becomes smaller if evading behavior has already been considered. This is due to the cost of the detecting method. When we introduce the first avoiding behavior

**Table 2:** Average shoulder widths and bust depths of the male and female participants. The values in the parentheses represent population standard deviations. The average shoulder widths of the males and the females were significantly different (p < 0.05), whereas the average bust depths were not. Note that the average shoulder width and bust depth of the male participants are larger than the statistical values introduced in Sec. 4.3. This is likely due to clothing thickness.

Gender	Shoulder width [cm]	Bust depth [cm]
Male	48.4(3.7)	26.0(4.2)
Female	41.0(1.4)	23.4(2.4)

(either evasion or rotation), we need to implement both the detecting and avoiding methods. However, in the introduction of the second avoiding behavior, we do not need to add the detecting method anymore. Therefore, the cost of introducing bodyrotation behavior is not very high compared with the improvement of the accuracy if some interactions, which accompany the detecting method of other pedestrians, have already been considered in simulation models.

# 7. Setup and conditions of corridor experiment 2

In corridor experiment 1 described in Secs. 3 and 4, all four participants were male adolescents. Thus, the same experiment was performed once again with male and female participants of various ages to demonstrate that the obtained results are valid for all pedestrians.

The experiment was conducted in the lecture hall at RCAST, The University of Tokyo, Japan. Four courses were constructed using cardboard boxes. A single course was the same as that in corridor experiment 1 (Fig. 3a). There were five male participants, ages 21, 24, 29, 66, and 68, and four female participants, ages 36, 45, 61, and 64. The average shoulder widths and bust depths of the male and female participants are summarized in Tab. 2. The average shoulder widths of the males and females were significantly different in terms of statistics (p < 0.05), while the bust depths were not. The tablets for recording angular velocities were strapped on the participants. We instructed the participants similarly as we did in corridor experiment 1.

The width W of the two courses were varied as 60, 70, 80, 90, 100, 120, and 140 cm and the angular velocity of each participant was observed. By changing the passing side (2 patterns) and participant pairing (combination choosing 2 participants from 9 = 36 patterns), 72 types of trials were executed for each W. Thus,  $72 \times 7$  (widths) = 504 datasets were obtained in total.

# 8. Analysis of passing rotational angles in corridor experiment 2

#### 8.1. Data processing

The time series of the angular velocities  $\omega_i(t)$   $(i = 0, 1, 2, \cdots, \text{and 8}$  (ID number)) were obtained at approximately 50 Hz from the tablets. Then, the same data processing as in corridor experiment 1 was conducted. The datasets of the angular velocities  $\omega_i(t)$  were numerically integrated and the time series of the rotational angle of each participant  $\varphi_i(t) \in [-180^\circ, 180^\circ]$  was obtained. The time series of the pair that includes  $\delta t_{i,n} \geq$ 0.1 s were discarded since it was impossible to compute the accurate rotational angles for such time series, as explained in Sec. 4.1. Actually, 4, 3, 5, 3, 2, 4, and 4 datasets were removed from W = 60, 70, 80, 90, 100, 120, and 140 cm, respectively. The passing rotational angles  $\theta_i$  for W = 60, 70, 80, 90, and 100 cm were calculated in a manner similar to that employed for corridor experiment 1, since no clear body rotations were observed for W = 120 and 140 cm. Any dataset that does not satisfy the conditions of the two thresholds ( $\theta_i$  and  $\Delta t_{|\varphi|_{\text{max}}}$ ) is converted appropriately before being used for analysis, as described in Appendix B.

Figure 14a shows the pairs of passing rotational angles  $(\theta_i, \theta_j)$   $(i \neq j)$  in the experimental data after conversion. The passing rotational angles can be observed to increase with decreasing corridor width, as was also observed in corridor experiment 1.

#### 8.2. Validation of elliptic pedestrians

To validate elliptic pedestrians and assumption (3) during passing, regardless of gender and age, a calculation similar to that for corridor experiment 1 was conducted. Since the shoulder width and bust depth of each participant was measured, error parameters for the semi-major axis  $\epsilon_a$  and semi-minor axis  $\epsilon_b$  were introduced to minimize the following function (the least squares method):

$$g(\epsilon_a, \epsilon_b) = \sum_{\text{all data}} \left[ W - d_i(\theta_i, a_i + \epsilon_a, b_i + \epsilon_b) - d_j(\theta_j, a_j + \epsilon_a, b_j + \epsilon_b) \right]^2, (17)$$

where  $a_i$   $(a_j)$  and  $b_i$   $(b_j)$  are the half of the shoulder width (semi-major axis) and bust depth (semi-minor axis) of participant i (j), respectively. As a result,  $(\epsilon_a, \epsilon_b) = (1.6 \text{ cm}, 2.1 \text{ cm})$  were obtained, which are positive small values. Thus, it can be considered that  $a_i + \epsilon_a$   $(a_j + \epsilon_a)$  and  $b_i + \epsilon_b$   $(b_j + \epsilon_b)$  represent the effective shoulder width and bust depth, respectively, as the participants did not come in contact with each other throughout the experiment. Hence, the elliptic representation of pedestrians and assumption (3) are validated for both males and females of various ages. Furthermore, as  $\epsilon_a < \epsilon_b$ , it was reconfirmed that pedestrians hesitate to a greater extent to come into contact with other pedestrians at their chest than at their shoulders.

The mean and sample standard deviation of  $d_i + d_j$  were computed for each W by substituting the half of the shoulder width and bust depth  $(a_i, b_i)$ ,  $(a_j, b_j)$ , calibrated parameters  $(\epsilon_a, \epsilon_b)$ , and experimental passing rotational angles  $(\theta_i, \theta_j)$  into (1). Figure 14b shows the mean and sample standard deviation of  $d_i + d_j$  with the red markers

Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 25



Figure 14: (a) Pairs of passing rotational angles  $(\theta_i, \theta_j)$  obtained from the experimental data after conversion for various values of W. The black crosses (+), red crosses (×), green triangles ( $\Delta$ ), blue circles ( $\circ$ ), and gray squares ( $\Box$ ) represent the experimental data for W = 60, 70, 80, 90, and 100 cm, respectively. Note that the experimental data are plotted such that  $\theta_i \geq \theta_j$  is satisfied. (b) Sum of the widths of the two ellipses  $d_i + d_j$  as a function of corridor width W. The green diagonal line represents theoretical assumption (3), i.e.,  $d_i + d_j = W$ . The red markers are the average values calculated using the least squares method by exploiting the experimental data and (1). The error bars show the sample standard deviations.

and error bars, respectively. The correspondence between the red markers and green diagonal line also indicates the validity of this assumption. Some discrepancy between the red marker and green diagonal line can be observed at W = 100 cm. As the female participants, whose average shoulder width was 41.0 cm as shown in Tab. 2, were included in the experiment, the value of  $d_i + d_j$  decreases.

#### 8.3. Difference between the male and female pairs

Since the average shoulder widths of the male and female participants were significantly different, the male and female pairs were analyzed separately in this section  $\parallel$ . Figure 15a shows the average passing rotational angles  $\theta_{ave}$  as functions of corridor width W. The plots with the blue squares and red circles are the results of the male and female pairs, respectively. The results of the male-female pairs, which were included in Fig. 14, were removed from Figure 15. It can be seen that  $\theta_{ave}$  decreases as W increases for both the male and female pairs. Moreover,  $\theta_{ave}$  is significantly larger for the males than

<sup>||</sup> There were no clear differences in the body sizes or passing rotational angles between the younger (adolescents and middle-aged adults) and older (elderlies) participants.

Body-rotation behavior of pedestrians for collision avoidance in passing and cross flow 26



Figure 15: (a) Average passing rotational angles as functions of corridor width W for the male and female pairs. Blue squares and red circles represent the result of the male pairs and the female pairs, respectively. The error bars show the sample standard deviations. We see that the average rotational angles for the male pairs are significantly larger than those for the female pairs at W = 60 and 70 cm (p < 0.05). (b) Sum of widths of the two ellipses  $d_i + d_j$  as functions of corridor width W for the male and female pairs. Blue squares and red circles represent the result of the male pairs and the female pairs, respectively. The error bars show the sample standard deviations. The green diagonal line represents theoretical assumption (3), that is,  $d_i + d_j = W$ . We see that the results of both the male and female pairs agree well with theoretical assumption (3) when the corridor width W is smaller than the double of the average shoulder widths described in Tab. 2.

for the females at W = 60 and 70 cm (p < 0.05), respectively. These differences can be attributed to the difference in the average shoulder width. Since the shoulder widths of the males were significantly larger than those of the females, the males had to rotate more than the females.

The error parameters for the male  $(\epsilon_{a,M}, \epsilon_{b,M})$  and female  $(\epsilon_{a,F}, \epsilon_{b,F})$  pairs were also computed. For the male pairs, the data for W = 60-100 cm were used and  $(\epsilon_{a,M}, \epsilon_{b,M})$ were calculated, which minimized (17) with the least square method. For the female pairs, only the data for W = 60-80 cm were used as the females did not need to rotate for W = 90 and 100 cm; this is because the maximum sum of the shoulder width of the pair was smaller than 90 cm. It was found that  $(\epsilon_{a,M}, \epsilon_{b,M}) = (0.0 \text{ cm}, 1.9 \text{ cm})$ and  $(\epsilon_{a,F}, \epsilon_{b,F}) = (1.2 \text{ cm}, 3.0 \text{ cm})$ ; these values were small enough to be regarded as the difference between the effective and real body sizes.  $\epsilon_{a,M} < \epsilon_{b,M}$  and  $\epsilon_{a,F} < \epsilon_{b,F}$ , confirming once again that pedestrians hesitate to make chest contact with other pedestrians more so than shoulder contact. Furthermore, since  $\epsilon_{a,M} < \epsilon_{a,F}$  and  $\epsilon_{b,M} < \epsilon_{b,F}$ , it can be concluded that the females kept a greater distance between each other than the males in the passing situation in our experiment.

Figure 15b shows the sum of the widths of the two ellipses  $d_i + d_j$  as functions of corridor width W for the male and female pairs. It can be observed that the results for the male pairs agree with the theoretical assumption (3) for W = 60-90 cm. At W = 100 cm, the result for the male pairs is smaller than the theoretical assumption (3). This small discrepancy can be explained as follows: since the average shoulder width of the males was 48.4 cm, as shown in Tab. 2, the males did not need to rotate for W = 100 cm in most cases; however, pedestrians always rotated to some degree during walking (see Appendix B). Thus, the widths of the males reduced to below 100 cm, and the blue square was plotted below the green diagonal line at W = 100 cm. On the other hand, the results of the females agree with the theoretical assumption (3) for W = 60-80 cm. As we mentioned in the second paragraph in this section, the maximum sum of the shoulder width of the female pairs was smaller than 90 cm, so that the discrepancies at W = 90 and 100 cm do not harm our assumption.

#### 9. Cross flow experiment

This section considers cross flow and investigates body-rotation behavior in a more complex and realistic scenario. There are two motivations for studying cross flow in this paper.

The first motivation is to observe that pedestrians do step sideways, i.e., discrepancies between the body orientation and walking directions in cross flow. Sidestepping was observed in the bidirectional-flow experiment in [40]. If such a phenomenon is observed in cross flow, which is more realistic than passage in a narrow corridor and more complex than bidirectional flow, it can be strongly confirmed once again that body rotation is an indispensable element in simulating pedestrian dynamics.

The second motivation is to understand the relationship between density and body rotation. A narrow corridor necessitates body rotation; however, there are several options to avoid collisions in cross flow. Although these additional options are beyond the scope of this paper, clarifying the relationship between density and body rotation would show when body rotation behavior becomes important in simulations.

The experimental setup and conditions are discussed in Sec. 9.1, and the results that correspond to the two motivations presented above are discussed in Secs. 9.2 and 9.3, respectively.

#### 9.1. Setup and conditions

The experiment was conducted on the street in front of RCAST building 4 at The University of Tokyo, Japan. An intersection, like the one shown in Fig. 16, was constructed by creating four paths intersecting each other in a delimited area. A video camera placed in an azimuthal position 20 m above the ground was used to record



Figure 16: Geometrical configuration for the cross-flow experiment (at Execution 1). South leg had to be bent due to space limitations. Starting positions are given as dashed lines along the four legs. Participants were allowed to take any position along the line, but only one participant was allowed at each line. Configuration given here is only representative as participants took different positions during each repetition.

pedestrian motion in the intersecting section. Due to space limitations, South leg had to be bent as shown in Fig. 16. Note that later data analysis (in particular concerning the flow in the four different directions [13]) revealed that the bending did not influence on participants' behaviors, making it possible to consider each leg independently.

Inside each leg, starting lines were drawn every 1 m longitudinally on the ground (the first starting line was 4 m from the center of the intersection). The participants were allowed to take any position along the different starting lines, but only one participant was allowed per line.

In total, 50 male college students took part in the experiment and were divided into the four legs in groups of 12 or 13 participants (Group A, B, C, and D in Tab. 3). As a consequence, unaligned lanes of respectively 11 and 12 m formed in the four legs. An ID number was assigned to each participant and each one was asked to turn in a different direction at the intersection (details are given in Tab. 3). An almost equal portion of participants had to turn left and right. Participants going straight were less than the ones turning. The turning direction was given them using a small paper provided at the beginning of the experiment; therefore, no one was aware of the turning direction

	Group	Turning right	Turning left	Going straight
-	А	$1,\!4,\!7,\!10$	$2,\!5,\!8,\!11$	$3,\!6,\!9,\!12$
-	В	$13,\!15,\!18,\!21,\!23$	14, 16, 19, 22, <b>24</b>	17,20,25
-	С	26,28,31, <b>35</b> ,37	27,29,32, <b>36</b>	30,33,34
-	D	39,41,44, <b>47</b> ,50	38,40,42,45, <b>48</b>	43,46,49
-	Total	19 (4  tablets)	18 (4  tablets)	13 (2  tablets)

**Table 3:** Group and turning direction assigned to each participant. Participants equippedwith a tablet are given in bold.

Table 4: Starting leg (West, South, North, East) for each group (A, B, C, D) during the different executions. This strategy was chosen to avoid a possible learning resulting from the repeated executions. Cardinal directions refer to Fig. 16. (The same group locations are relative to Execution 1.)

Execution	West	South	North	East
1	А	В	С	D
2	А	С	В	D
3	D	С	В	А
4	D	В	С	А

of others.

The participants started walking simultaneously toward the intersection when instructed to go. Then, they turned according to the assigned directions at the intersection. After leaving the intersection, the participants walked for a long distance to avoid the formation of congestion close to the exit.

To avoid any possible learning process, in other words, to repeat the experiment under "fresh" conditions, the groups were switched after each repetition, as shown in Tab. 4. This switching strategy made it very unlikely that participants were able to memorize others' behaviors. The participants' positions were also longitudinally switched at each repetition to avoid always having the same participants at the beginning or end of each leg.

The participants were given caps of different colors allowing us to precisely track their positions. Moreover, as shown in Tab. 3, some of the participants (10 in total) were equipped with a tablet to measure their bodies' rotational angles, as discussed in Sec. 3. The participants equipped with tablets wore blue, red or green caps, whereas the rest of the participants wore yellow caps. By knowing the combination of cap's color and starting leg, we were able to combine the body rotational angles and the positions (or trajectories) of each participant obtained from the tablets and the video camera, respectively. This allowed the time at which a given participant crossed the intersection to be precisely determined and the data relative to that particular moment to be analyzed.

Finally, to check the impact of density on body rotation, three different widths



Figure 17: Configuration of the central part of the intersection for the leg width  $W_{\text{leg}} = 3$ , 2, and 1 m. Width of the waiting areas in each legs were kept unchanged, while a bottleneck was created in each entrance to increase the density in the central part.

 $(W_{\text{leg}} = 3, 2, \text{ and } 1 \text{ m})$  were investigated for the intersecting legs. For this purpose, the final parts of the four legs were restricted, creating a bottleneck before entering the intersection. As shown in Fig. 17, a particular setup was chosen to realize a straight path before the intersection in each configuration. Note that the starting positions (the lines of Fig. 16) and experimental concepts described so far were unchanged. Video frames relative to each configuration are shown in Fig. 18. Inaccessible areas created from changing the leg width were colored white to focus on the experimental area.

Overall, the presented experimental setup allowed for the recreation of a fairly natural cross flow in which participants had to consider each situation independently. The careful planning concerning turning directions and starting positions enabled comparable situations, in which only density played an important role, to be considered.

Since it was difficult to distinguish turning and body rotation clearly for participants turning right and left at the intersection, the focus was placed on the participants equipped with the tablets and going straight, i.e., the participants 3 and 49 (Tab. 3) in the following analyses. The analyses of the other participants are presented in [13].



Figure 18: Video frames from the three different configurations. (a)  $W_{\text{leg}} = 3 \text{ m}$ , (b)  $W_{\text{leg}} = 2 \text{ m}$ , and (c)  $W_{\text{leg}} = 1 \text{ m}$ . To focus on the experimental area, inaccessible areas were cut from the images. For each configuration, four executions were performed with the different initial conditions described in Tab. 4.

# 9.2. Discrepancy between body rotational angle and walking direction

Figure 19a shows the snapshot of participant 3 with a blue cap walking from the West (Left) to the East (Right) in the case  $W_{\text{leg}} = 2$  m. It can be seen that he avoided other participants by rotating his body counter-clockwise and tried to move to the right. This phenomenon is understood more clearly by analyzing the evolution of the body rotational angle  $\varphi$  and walking direction  $\alpha$ , as in Fig. 19b. The body rotational angle  $\varphi$  becomes large at t = 4.5 s, and a great discrepancy was observed between the body rotational angle  $\varphi$  and walking direction  $\alpha$ . These results correspond well to the snapshot (Fig. 19a), i.e., participant 3 successfully moved to the right without changing walking direction by using body rotation to avoid others. This is an experimental fact that a pedestrian avoids others by rotating his body without greatly changing his walking direction.

Figures 19c and 19d show similar results. Participant 49 with a red cap walked from the West (Left) to the East (Right) in the case  $W_{\text{leg}} = 2$  m. Here, he rotated his body clockwise and tried to walk through other participants. Evolution in the walking direction indicates that participant 49 walked to the south-east direction with some degree, since the absolute walking direction achieved  $|\alpha| \approx 20$  deg. However, the absolute body rotational angle  $|\varphi|$  achieved at t = 10.8 s was much larger. Therefore, participant 49 is considered to have rotated his body not only for changing his walking direction, but also for going through other participants.

The discrepancies between the body rotational angles (body orientation) and walking directions indicate that they must be dealt with separately.

### 9.3. Relation between density and body rotational angle

Next, the relationship between the density and body rotational angle were investigated. Figure 20 shows the body rotational angle  $\varphi$  as a function of the local density  $\rho$  for participants 3 and 49. The local densities  $\rho$  were calculated by depicting the Voronoi





Figure 19: Snapshots and evolution of body rotational angle  $\varphi$  and walking direction  $\alpha$  in the cross flow experiment for  $W_{\text{leg}} = 2$  m. The snapshots correspond to the times when the maximum absolute rotational angles  $|\varphi|_{\text{max}}$  were achieved. (a), (b) The results of participant 3 with a blue cap walking from the West (Left) to the East (Right) in Execution 1. (c), (d) The results of participant 49 with a red cap walking from the West (Left) to the East (Right) in Execution 3.

diagram using the video camera images. The blue, green, and red plots are the instantaneous results for  $W_{\text{leg}} = 3$ , 2, and 1 m, respectively. Since the frame rate of the video camera (29.97 frame/sec (NTSC)) was lower than the frequency of the tablets (50 Hz), the resolution of the data was reduced to that of the video camera ¶.

<sup>¶</sup> Note that the successive data points are strongly correlated; however, this correlation does not harm our result. Since the time of one trial was long enough, we could observe various situations in one trial, in other words, the data are not biased toward specific situations. If we decrease the sampling rate, the plots become sparse. Furthermore, the maximum rotational angles may change because we sometimes fail to observe the larger rotational angles due to the low sampling rate. Therefore, high sampling rate is important to see the body rotation by analyzing rotational angles. However, we confirmed that the changes of the maximum rotational angles were small, i.e, the positions of the black circles, squares, and triangles in Fig. 20 do not greatly change by the sampling rate, due to the strong correlations



Figure 20: Body rotational angle  $\varphi$  as a function of the local density  $\rho$  for participants 3 and 49. The blue dots (·), green crosses (+), and red crosses (×) represent the instantaneous results for  $W_{\text{leg}} = 3$ , 2, and 1 m, respectively (29.97 frame/sec (NTSC)). The black circles, squares, and triangles show the maximum absolute rotational angles in each trial for  $W_{\text{leg}} = 3$ , 2, and 1 m, respectively. Note that the data, which corresponds to the situations in Figs. 19ab and 19cd, are also plotted in the figure.

Note that the data from the tablets were interpolated to depict the figure. The black circles, squares, and triangles show the maximum absolute rotational angles in each trial for  $W_{\text{leg}} = 3$ , 2, and 1 m, respectively.

The body rotational angles were observed to achieve  $\varphi \approx 20$  deg for a broad range of densities. In the cross flow, the participants needed to avoid others from multiple directions, thus, they rotated their body to some degree even at low densities. Furthermore, large body rotational angles were achieved in only the low and middle density regions ( $\rho < 2.5 \text{ m}^{-2}$ ). When the density was not high, a participant could look for a space between the others and went through there by rotating his body. It was confirmed that the large body rotational angles  $\varphi$  observed for  $\rho < 2.5 \text{ m}^{-2}$  corresponded to such phenomenon. However, for higher densities ( $\rho > 2.5 \text{ m}^{-2}$ ), it was difficult for the participants to find enough space to go through. Thus, they slowed down or stopped and waited until enough space appeared in front of them. Hence, no large body rotational angles  $\varphi$  could be seen in the high-density region.

As Fig. 2 shows, the decrease in the width of an ellipse is small for small rotational angles. Therefore, the conventional models can simulate cross flow well to some extent, even if they do not include body-rotation behavior, since most pedestrians do not rotate greatly. However, such models cannot reproduce pedestrians going through others by body rotation, who were observed in our cross-flow experiment. Hence, body-rotation behavior is indeed an indispensable factor for more realistic simulations.

between the successive data.

# 10. Conclusion

This study mainly investigated pedestrians' collision avoidance behaviors involving body rotation and side-stepping, when body orientation differs from walking direction, such as in a narrow corridor. A pedestrian was modeled with an ellipse, and the mathematical relationship between the body rotational angles of two elliptic pedestrians and the corridor width was obtained. Moreover, experiments with real pedestrians were conducted, and the rotational angles were measured using commercial tablets. The results of the experiment indicate that pedestrians start to rotate when their shoulders may come into contact, and the rotational angles during passing increase as the corridor width decreases. Finally, the sizes of the elliptic pedestrians were determined by the least squares method, using both the theoretical and experimental results. The calibrated size of the elliptic pedestrians quantitatively agreed with the effective size of real pedestrians. We would like to emphasize that the participants of our experiments included both males and females with various ages, so that we consider that our results are not limited to the specific types of pedestrians. Furthermore, a simulation model for pedestrians passing through a narrow corridor was developed. The simulation results agreed with the experimental results.

A cross-flow experiment was conducted to investigate body-rotation behavior in a more complex and realistic scenario. The results indicated the importance of considering body-rotation behavior in complex scenarios. It was also found that the participants tended to rotate greatly and stepped sideways when the density was not high. When the density was high, the participants could not move forward even with physical rotations, and thus large body rotations were not observed.

In the future, the combination of position and rotational angle data will allow for the analysis and modeling of when and how pedestrians start body rotation. The coupling effect of changing walking direction and body orientation on collision avoidance will also be studied.

# Acknowledgments

We would like to thank the entire team of the Nishinari group for their assistance during the planning and execution of the experiment. We also appreciate Takahiro Ezaki for his useful comments regarding our manuscript. The authors would like to thank Enago (www.enago.jp) for the English language review. This work was supported by JSPS KAKENHI Grant Numbers 25287026 and 15K17583, and JST-Mirai Program Grant Number JPMJMI17D4, Japan.

# Appendix A. Width of a rotated ellipse

Here, we provide details on derivation of the relationship between the passing rotational angle  $\theta \in [0, \pi/2]$  and the width d of an ellipse. With proper choice of coordinate system,

ellipse i may be expressed quadratically as follows:

$$\frac{x^2}{a^2} + \frac{y^2}{b^2} = 1, (A.1)$$

where a and b are the semi-major and the semi-minor axes of the ellipse, that is, a > b.



Figure A1: Schematic of rotated ellipse i'. Geometric relationship between rotational angle  $\theta$  and width d is also shown in the figure.

Next, we rotate ellipse *i* counter-clockwise by  $\theta$  with respect to its center O and consider ellipse *i'*, as in Fig. A1. Using rotational transformation, a point  $(x_2, y_2)$  on ellipse *i'*, which corresponds to a point  $(x_1, y_1)$  on ellipse *i*, is described as

$$\begin{pmatrix} x_2 \\ y_2 \end{pmatrix} = \begin{pmatrix} \cos\theta & -\sin\theta \\ \sin\theta & \cos\theta \end{pmatrix} \begin{pmatrix} x_1 \\ y_1 \end{pmatrix}.$$
 (A.2)

By deforming (A.2), we obtain

$$\begin{cases} x_1 = x_2 \cos \theta + y_2 \sin \theta, \\ y_1 = -x_2 \sin \theta + y_2 \cos \theta. \end{cases}$$
(A.3)

Since the point  $(x_1, y_1)$  is on ellipse *i*, it satisfies (A.1). Thus, we substitute (A.3) into (A.1) and obtain the equation of the rotated ellipse *i'* as follows:

$$\frac{(x\cos\theta + y\sin\theta)^2}{a^2} + \frac{(-x\sin\theta + y\cos\theta)^2}{b^2} = 1,$$
(A.4)

where we replace  $x_2$  and  $y_2$  with x and y, respectively.

We denote the tangents of ellipse i' that are parallel to the y axis as  $l_1$  and  $l_2$ , as in Fig. A1. Their equations are given as

$$\begin{cases} l_1 : x = d/2, \\ l_2 : x = -d/2, \end{cases}$$
(A.5)

because ellipse i' is symmetric with respect to its center O.

Then, we consider the intersection equation of ellipse i' and tangent  $l_1$  ( $l_2$ ) by substituting (A.5) into (A.4), which yields a quadratic equation with respect to y,

$$\frac{(d\cos\theta/2 + y\sin\theta)^2}{a^2} + \frac{(-d\sin\theta/2 + y\cos\theta)^2}{b^2} = 1.$$
 (A.6)

Since  $l_1$  ( $l_2$ ) is tangential to ellipse i', the discriminant of (A.6) D = 0. By solving this equation, we finally obtain the relation between  $\theta$  and d as follows:

$$d(\theta) = 2\sqrt{a^2 \cos^2 \theta + b^2 \sin^2 \theta}.$$
(A.7)

Note that more complicated scenarios, for instance, consideration of the distance between ellipses and lines, have been analyzed in other studies [48, 64, 65].

# Appendix B. Conversion method for passing rotational angle

In this appendix, we explain how we converted inappropriate passing rotational angles to appropriate ones. We decided not to simply remove but include them for analysis by conversion because removing such data, whose values are small, puts improper focus on rare data with clear rotation and awkwardly increases the average passing rotational angles.

As described in Sec. 4.2, we firstly determined the two thresholds: one for the passing rotational angle  $\theta_i$  and the other for the time gap between the points at which the participants achieved the maximum absolute rotational angle  $\Delta t_{|\varphi|_{\text{max}}}$ . The threshold for  $\theta_i$  was calculated for each participant by averaging the maximal absolute rotational angles during passing in W = 120 and 140 cm. In corridor experiment 1 (Secs. 3 and 4),  $\theta_{\text{th},1} = 8.4^{\circ}$ ,  $\theta_{\text{th},2} = 8.5^{\circ}$ ,  $\theta_{\text{th},3} = 6.0^{\circ}$ , and  $\theta_{\text{th},4} = 8.1^{\circ}$ . In corridor experiment 2 (Secs. 7 and 8),  $\theta_{\text{th},0} = 10.8^{\circ}$ ,  $\theta_{\text{th},1} = 8.7^{\circ}$ ,  $\theta_{\text{th},2} = 9.7^{\circ}$ ,  $\theta_{\text{th},3} = 6.8^{\circ}$ ,  $\theta_{\text{th},4} = 7.5^{\circ}$ ,  $\theta_{\text{th},5} = 8.2^{\circ}$ ,  $\theta_{\text{th},6} = 10.1^{\circ}$ ,  $\theta_{\text{th},7} = 7.4^{\circ}$ , and  $\theta_{\text{th},8} = 8.6^{\circ}$ . The threshold  $\theta_{\text{th},i}$  represents the maximum rotational (oscillation) angle in the pass without clear rotation of participant *i*. The threshold for  $\Delta t_{|\varphi|_{\text{max}}}$  was determined to be 0.5 s from Figs. 5 and 6.

Next, we applied the following procedure to all datasets corresponding to  $W \leq 100$  cm:

- (i) For a datum with  $|\varphi|_{\max,i} < \theta_{\text{th},i}$ , we judged that there was no clear body rotation and used  $\theta_{\text{th},i}$  as a passing rotational angle <sup>+</sup>.
- (ii) For a dataset with  $\Delta t_{|\varphi|_{\max}} > 0.5 \text{ s}$ ,  $|\varphi|_{\max,i} > \theta_{\text{th},i}$  and  $|\varphi|_{\max,j} > \theta_{\text{th},j}$  ( $|\varphi|_{\max,i} > |\varphi|_{\max,i}$ ), we judged that passing occurs at  $t_{|\varphi|_{\max,i}}$ , that is, the moment when  $|\varphi|_{\max,i}$  was achieved. Then, we determined passing rotational angles  $\theta_i = |\varphi|_{\max,i}$  and  $\theta_j$  as the larger of  $\theta_{\text{th},j}$  and the maximum rotational angle achieved between  $t_{|\varphi|_{\max,i}} 0.5 \text{ s}$  and  $t_{|\varphi|_{\max,i}} + 0.5 \text{ s}$ .

<sup>&</sup>lt;sup>+</sup> As shown in Figs. 5g and 5h, we observed some oscillations of the rotational angles even though there were no clear body rotations. Therefore, if we used  $0^{\circ}$  as the passing rotational angles in the cases of no clear body rotation, downward bias would arise in the average passing rotational angles. Hence,  $\theta_{\text{th},i}$  was adopted as the passing rotational angle instead of  $0^{\circ}$ .

In corridor experiment 1, step (i) was applied to 3 and 17 data for W = 90 and 100 cm, respectively. Step (ii) was applied to 2 and 7 data for W = 90 and 100 cm, respectively. In corridor experiment 2, step (i) was applied to 8 and 18 data for W = 90 and 100 cm, respectively. Step (ii) was applied to 10, 4, 6, 38, and 62 data for W = 60, 70, 80, 90 and 100 cm, respectively. Note that 48 and 144 data (24 and 72 datasets) were obtained for each corridor width W in corridor experiment 1 and 2 as described in Secs. 3 and 7, respectively.

# References

- F Schweitzer. Browning Agents and Active Particles. Springer Series in Synergetics. Springer Berlin Heidelberg, Berlin, Heidelberg, 2007.
- [2] D Helbing. Traffic and related self-driven many-particle systems. Rev. Mod. Phys., 73(4):1067– 1141, 2001.
- [3] A Schadschneider, D Chowdhury, K Nishinari, and L Santen. Stochastic Transport in Complex Systems. Elsevier, Amsterdam/Oxford, 2010.
- [4] D Helbing and P Molnár. Social force model for pedestrian dynamics. Phys. Rev. E, 51(5):4282–4286, 1995.
- [5] D Helbing, I Farkas, and T Vicsek. Simulating dynamical features of escape panic. Nature, 407(6803):487–490, 2000.
- [6] C Burstedde, K Klauck, A Schadschneider, and J Zittartz. Simulation of pedestrian dynamics using a two-dimensional cellular automaton. *Physica A*, 295(3-4):507–525, 2001.
- [7] A Seyfried, B Steffen, W Klingsch, and M Boltes. The fundamental diagram of pedestrian movement revisited. J. Stat. Mech. Theory Exp., 2005(10):P10002–P10002, 2005.
- [8] A Jelić, C Appert-Rolland, S Lemercier, and J Pettré. Properties of pedestrians walking in line: Fundamental diagrams. *Phys. Rev. E*, 85(3):036111, 2012.
- [9] S. P Hoogendoorn and W Daamen. Self-Organization in Pedestrian Flow. In Traffic and Granular Flow '03, pages 373–382. Springer-Verlag Berlin Heidelberg, 2005.
- [10] T Kretz, A Grünebohm, M Kaufman, F Mazur, M Schreckenberg, A Grunebohm, M Kaufman, F Mazur, and M Schreckenberg. Experimental study of pedestrian counterflow in a corridor. J. Stat. Mech. Exp., 10001(10):P10001, 2006.
- [11] J Zhang, W Klingsch, A Schadschneider, and A Seyfried. Ordering in bidirectional pedestrian flows and its influence on the fundamental diagram. J. Stat. Mech. Theory Exp., 2012(02):P02002, 2012.
- [12] C Feliciani and K Nishinari. Empirical analysis of the lane formation process in bidirectional pedestrian flow. *Phys. Rev. E*, 94(3):032304, 2016.
- [13] C Feliciani. Measurement and numerical modeling of pedestrian flows. PhD dissertation, The University of Tokyo.
- [14] S. C Wong, W. L Leung, S. H Chan, W. H. K Lam, N. H. C Yung, C. Y Liu, and P Zhang. Bidirectional Pedestrian Stream Model with Oblique Intersecting Angle. J. Transp. Eng., 136(3):234-242, 2010.
- [15] F. S Hänseler, W. H Lam, M Bierlaire, G Lederrey, and M Nikolić A dynamic network loading model for anisotropic and congested pedestrian flows. *Transp. Res. Part B Methodol.*, 95:149– 168, 2017.
- [16] S. P Hoogendoorn and W Daamen. Pedestrian Behavior at Bottlenecks. Transp. Sci., 39(2):147– 159, 2005.
- [17] T Kretz, A Grünebohm, and M Schreckenberg. Experimental study of pedestrian flow through a bottleneck. J. Stat. Mech. Theory Exp., 2006(10):P10014–P10014, 2006.

- [18] A Seyfried, O Passon, B Steffen, M Boltes, T Rupprecht, and W Klingsch. New Insights into Pedestrian Flow Through Bottlenecks. *Transp. Sci.*, 43(3):395–406, 2009.
- [19] A Seyfried, B Steffen, and T Lippert. Basics of modelling the pedestrian flow. Physica A, 368(1):232–238, 2006.
- [20] D. R Parisi, M Gilman, and H Moldovan. A modification of the Social Force Model can reproduce experimental data of pedestrian flows in normal conditions. *Physica A*, 388(17):3600–3608, 2009.
- [21] A Kirchner, H Klüpfel, K Nishinari, A Schadschneider, and M Schreckenberg. Simulation of competitive egress behavior: comparison with aircraft evacuation data. *Physica A*, 324(3-4):689– 697, 2003.
- [22] A Kirchner, K Nishinari, and A Schadschneider. Friction effects and clogging in a cellular automaton model for pedestrian dynamics. *Phys. Rev. E*, 67(5):056122, 2003.
- [23] D Yanagisawa, A Kimura, A Tomoeda, R Nishi, Y Suma, K Ohtsuka, and K Nishinari. Introduction of frictional and turning function for pedestrian outflow with an obstacle. *Phys. Rev. E*, 80(3):036110, 2009.
- [24] C. M Henein and T White. Macroscopic effects of microscopic forces between agents in crowd models. *Physica A*, 373:694–712, 2007.
- [25] C. M Henein and T White. Microscopic information processing and communication in crowd dynamics. *Physica A*, 389(21):4636–4653, 2010.
- [26] Y Suma, D Yanagisawa, and K Nishinari. Anticipation effect in pedestrian dynamics: Modeling and experiments. *Physica A*, 391(1-2):248–263, 2012.
- [27] S Nowak and A Schadschneider. Quantitative analysis of pedestrian counterflow in a cellular automaton model. *Phys. Rev. E*, 85(6):066128, 2012.
- [28] W. G Weng, T Chen, H. Y Yuan, and W. C Fan. Cellular automaton simulation of pedestrian counter flow with different walk velocities. *Phys. Rev. E*, 74(3):036102, 2006.
- [29] R Jiang and Q.-S Wu. Pedestrian behaviors in a lattice gas model with large maximum velocity. *Physica A*, 373:683–693, 2007.
- [30] P Hrabák and M Bukáček. Conflict Solution According to Aggressiveness of Agents in Floor-Field-Based Model. Lect. Notes Comput. Sci. (Parallel Process. Appl. Math.), 3911:507–516, 2016.
- [31] S Bandini, S Manzoni, and G Vizzari. Agent Based Modeling and Simulation : An Informatics Perspective. J. Artif. Soc. Soc. Simul., 12(4):4, 2009.
- [32] S Bandini, L Crociani, and G Vizzari. Pedestrian Simulation: Considering Elderlies in the Models and in the Simulation Results. In *Ambient Assist. Living*, pages 11–21. Springer International Publishing, 2015.
- [33] D Yanagisawa, A Tomoeda, and K Nishinari. Improvement of pedestrian flow by slow rhythm. *Phys. Rev. E*, 85(1):016111, 2012.
- [34] A Jelić, C Appert-Rolland, S Lemercier, and J Pettré. Properties of pedestrians walking in line. II. Stepping behavior. *Phys. Rev. E*, 86(4):046111, 2012.
- [35] I von Sivers and G Köster. Dynamic stride length adaptation according to utility and personal space. Transp. Res. Part B Methodol., 74:104–117, 2015.
- [36] X Liu, W Song, L Fu, W Lv, and Z Fang. Typical features of pedestrian spatial distribution in the inflow process. *Phys. Lett. A*, 380(17):1526–1534, 2016.
- [37] T Ezaki, K Ohtsuka, M Chraibi, M Boltes, D Yanagisawa, A Seyfried, A Schadschneider, and K Nishinari. Inflow Process of Pedestrians to a Confined Space. *Collect. Dyn.*, 1:A4, 2016.
- [38] T Ezaki, D Yanagisawa, K Ohtsuka, and K Nishinari. Simulation of space acquisition process of pedestrians using Proxemic Floor Field Model. *Physica A*, 391(1-2):291–299, 2012.
- [39] T Ezaki, K Ohtsuka, D Yanagisawa, and K Nishinari. Inflow Process: A Counterpart of Evacuation. In M Chraibi, M Boltes, A Schadschneider, and A Seyfried, editors, *Traffic and Granular Flow '13*, pages 227–231. Springer International Publishing, Cham, 2015.
- [40] C.-J Jin, R Jiang, J.-L Yin, L.-Y Dong, and D Li. Simulating bi-directional pedestrian flow in a cellular automaton model considering the body-turning behavior. *Physica A*, 482:666–681,

2017.

- [41] V. J Blue and J. L Adler. Cellular automata microsimulation for modeling bi-directional pedestrian walkways. Transp. Res. Part B Methodol., 35(3):293–312, 2001.
- [42] C Feliciani and K Nishinari. An improved Cellular Automata model to simulate the behavior of high density crowd and validation by experimental data. *Physica A*, 451:135–148, 2016.
- [43] D Yanagisawa. Coordination Game in Bidirectional Flow. Collect. Dyn., 1:A8, 2016.
- [44] K Nagao, D Yanagisawa, and K Nishinari. Estimation of crowd density applying wavelet transform and machine learning. *Physica A*, 510:145–163, 2018.
- [45] J. J Fruin. Pedestrian planning and design. Metropolitan Association of Urban Designers and Environmental Planners, 1971.
- [46] J Pauls. Suggestions on evacuation models and research questions. In T. J Shields, editor, Proc. Third Int. Symp. Hum. Behav. Fire, number 2001, London, 2004. Interscience.
- [47] J Was, B Gudowski, and P. J Matuszyk. Social Distances Model of Pedestrian Dynamics. Lect. Notes Comput. Sci. (Cellular Automata), 4173:492–501, 2006.
- [48] M Chraibi, A Seyfried, and A Schadschneider. Generalized centrifugal-force model for pedestrian dynamics. *Phys. Rev. E*, 82(4):046111, 2010.
- [49] M Moussaid, D Helbing, S Garnier, A Johansson, M Combe, and G Theraulaz. Experimental study of the behavioural mechanisms underlying self-organization in human crowds. Proc. R. Soc. B Biol. Sci., 276(1668):2755–2762, 2009.
- [50] T Imai, S. T Moore, T Raphan, and B Cohen. Interaction of the body, head, and eyes during walking and turning. *Exp. Brain Res.*, 136(1):1–18, 2001.
- [51] A Gorrini, K Shimura, S Bandini, K Ohtsuka, and K Nishinari. Experimental Investigation of Pedestrian Personal Space. Transp. Res. Rec. J. Transp. Res. Board, 2421:57–63, 2014.
- [52] A. H Olivier, A Marin, A Crétual, A Berthoz, and J Pettré. Collision avoidance between two walkers: role-dependent strategies. *Gait & Posture*, 38(4):751–756, 2013.
- [53] W Daamen, S Hoogendoorn, M Campanella, and D Versluis. Interaction Behavior Between Individual Pedestrians. *Pedestrian and Evacuation Dynamics 2012*, pages 1305–1313, 2014.
- [54] M Huber, Y.-H Su, M Krüger, K Faschian, S Glasauer, and J Hermsdörfer. Adjustments of Speed and Path when Avoiding Collisions with Another Pedestrian. *PLoS One*, 9(2):e89589, 2014.
- [55] D. R Parisi, P. A Negri, and L Bruno. Experimental characterization of collision avoidance in pedestrian dynamics. *Phys. Rev. E*, 94(2):022318, 2016.
- [56] F Zanlungo, T Ikeda, and T Kanda. A Microscopic "Social Norm" Model to Obtain Realistic Macroscopic Velocity and Density Pedestrian Distributions. *PLoS One*, 7(12):e50720, 2012.
- [57] C Feliciani and K Nishinari. Pedestrians rotation measurement in bidirectional streams. In W Song, J Ma, and L Fu, editors, *Pedestrian and Evacuation Dynamics 2016*, pages 76–83, Hefei, 2016. University of Science and Technology of China Press.
- [58] C Feliciani and K Nishinari. Estimation of pedestrian crowds' properties using commercial tablets and smartphones. *Transportmetrica B: Transport Dynamics* 2018. doi: 10.1080/21680566.2018.1517061 (published online)
- [59] Report on size-JPN 2004-2006. Ministry of Economy, Trade and Industry in Japan, 2007. (in Japanese) URL http://warp.da.ndl.go.jp/info:ndljp/pid/281883/www.meti.go.jp/press/ 20071001007/20071001007.html
- [60] S Holl. Methoden für die Bemessung der Leistungsfähigkeit multidirektional genutzter Fußverkehrsanlagen. PhD thesis, Universität Wuppertal, 2016. (in German) URL http://juser.fz-juelich.de/record/825757?ln=en
- [61] S Cao, A Seyfried, J Zhang, S Holl, and W Song. Fundamental diagrams for multidirectional pedestrian flows. J. Stat. Mech. Theory Exp., 2017(3), 2017.
- [62] Jülich Research Center. Data archive of experimental data from studies about pedestrian dynamics / Unidirectional flow. URL http://ped.fz-juelich.de/da/2013unidirectional
- [63] Jülich Research Center. Data archive of experimental data from studies about pedestrian dynamics / Bidirectional flow. URL http://ped.fz-juelich.de/da/2013bidirectional

- [64] X Zheng and P Palffy-Muhoray. Distance of closest approach of two arbitrary hard ellipses in two dimensions. *Phys. Rev. E*, 75(6):061709, 2007.
- [65] M Chraibi. Distance of closest approach of ellipses and ellipsoids. ResearchGate, 2012. URL https://www.researchgate.net/publication/ 306107046\_Distance\_of\_closest\_approach\_of\_ellipse\_and\_line\_segment